Author's response to reviews

Title: Outcomes of sustained low efficiency dialysis versus continuous renal replacement therapy in critically ill adults with acute kidney injury: A cohort study

Authors:

Abhijat Kitchlu (abhijat.kitchlu@gmail.com)
Neill Adhikari (neill.adhikari@sunnybrook.ca)
Karen E.A. Burns (burnsk@smh.ca)
Jan O. Friedrich (friedrichj@smh.ca)
Amit X. Garg (amit.garg@lhsc.on.ca)
David Klein (kleind@smh.ca)
Robert M. Richardson (robert.richardson@uhn.ca)
Ron Wald (waldr@smh.ca)

Version: 2
Date: 6 September 2014

Author's response to reviews:

Dear Ms. Lippman:

Thank you very much for your e-mail. We have submitted a revised version of the manuscript to the website as per the journal style requirements.

With regards to the invitation to submit previous reviewer comments, we would be happy to provide comments from our prior submission to Nephrology Dialysis Transplantation. We were very fortunate to receive thoughtful and valid comments from 3 reviewers on behalf of this journal. We have attached a Word document including these comments. As we did not re-submit to Nephrology Dialysis Transplantation we did not formulate direct responses to each of their comments. We did, however, do our best to address their concerns where possible in the revised manuscript that has now been submitted. We have also attached the original manuscript submitted to Nephrology Dialysis Transplantation for comparison. As the manuscript went through several iterations between the two submissions, unfortunately there is not a version of the document with all marked changes. We hope the comparison and reviewer comments will aid in the peer-review process for Critical Care. Please let us know if there is any additional information we can provide.

Sincere thanks,

Abhijat Kitchlu, MD
PGY-4, Adult Nephrology Fellowship Program